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Abstract 

The biopharma industry traditionally uses the batch model for manufacturing biotherapeutics 

due to risk tolerance and regulatory concerns. However, there is a growing interest in 

continuous processing to meet rising demand. Continuous processing offers benefits such as 

improved productivity, consistent product quality, and reduced costs. While the majority of 

research focus has been on the creation of end-to-end continuous biomanufacturing processes 

in recent years, limited attention has been paid to real time monitoring and control of continuous 

processes. The aim of this thesis is to develop model based designs and control of downstream 

purification steps along with development of appropriate online Process Analytical Technology 

(PAT) tool for real time monitoring of Critical Quality Attributes (CQAs) in continuous 

processing of biotherapeutics. 

The first objective of the work was to explore applications of machine learning (ML) algorithms 

for qualitative monitoring and quantitative prediction of column state and performance for resin 

cycling study of continuous capture chromatography. This process is crucial in monoclonal 

antibody purification due to resin costs. Advanced ML algorithms were proposed to detect 

column integrity breaches and predict yield decline in protein A chromatography. The 

developed algorithms successfully predicted anomalies by detecting subtle changes in column 

packing and decline in yield, enabling real-time control decisions for preventive measures. 

The second objective of the work was to model the complex resin decay phenomena due to 

extended re-use of protein A resin in continuous capture chromatography using general rate 

model (GRM) for monoclonal antibody (mAb). Our findings show that pore and film diffusion 

slow down with each cycle, affecting the protein transfer rate. It was also observed that although 

pore diffusion was the limiting step, film diffusion was also slowing with number of cycles 

leading to slower diffusion rate of protein out of solid phase to bulk phase. The methodology 

and findings proposed in this work can assist in predicting reliable performance of Protein A 

resin with useful insights regarding the underlying mechanism behind resin fouling.  

Besides process modeling, creation of appropriate PAT tool is required for real time monitoring 

of product quality attributes in continuous processing, thereby ensuring product quality and 

consistency. In this objective, a cost-effective approach for using a traditional offline high-

performance liquid chromatography (HPLC) for online analysis using a 2 way/6 port valve to 

facilitate simultaneous automated sampling of product stream eluting from a process column 

and fractionation was proposed. The versatility of the proposed online configuration has been 

verified by demonstrating its use for separation of aggregates and separation of charge variants 
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of mAb. Furthermore, predictive modeling was implemented to facilitate pooling to achieve 

the desired purity. The study demonstrates that the proposed online HPLC configuration could 

be used for PAT applications in continuous preparative chromatography. 

Finally, model based design and control of end-to-end integrated downstream purification 

process coupled with online HPLC PAT tool for achieving robust pooling decision making in 

continuous processing of mAb was proposed. Robust pooling decision is required for achieving 

uniform charge variant composition for mAbs. In this study, we proposed an end-to-end 

modeling of continuous downstream process providing for digital twin, integrated with an 

online HPLC-PAT tool for delivering real time pooling decisions to achieve uniform charge 

variant composition. The model states were updated in real time using online HPLC charge 

variant data for prediction of the initial and final cut point for cation exchange chromatography 

(CEX) eluate, according to which the process chromatography was directed to switch from 

collection to waste to achieve the desired charge variant composition in the CEX pool. The 

control strategy was successfully implemented for wide range of charge variants in the feed 

with its composition in the range of 13-17% for acidic, 18-23% for main and 59-68% for basic 

variants. All the case studies yielded CEX pool of uniform distribution of charge variants with 

process yield for main species was >85%. The results thus successfully demonstrate the 

technical feasibility of creating digital twins of bioprocess operations and their utility for 

process control. 

In order to run a reliable, long-term continuous process with increased efficiency and 

robustness, we believe the studies presented here as a part of this thesis will promote 

development of several modeling based techniques together with application of appropriate 

PAT tools for real time monitoring and control of bioprocess unit operations. 
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सार 

जोखिम सह्यता और विनियामकीय कारणों से बायोफामाा उद्योग परंपरागत रूप स े
बायोथेरेप्यूटिक्स के निमााण के लिए बैच मॉडि का उपयोग करता है। हािााँकक, बढ़ती मााँग को 
पूरा करिे के लिए अविरत प्रसंस्करण में रुचच बढ़ रही है। अविरत प्रसंस्करण से बेहतर 
उत्पादकता, सुसंगत उत्पाद गुणित्ता और कम िागत जैस ेिाभ लमिते हैं। जहा ंपर हाि के 
िर्षों में अिुसंधाि का अचधकांश ध्याि एंड-िू-एंड अविरत जैि-विनिमााण प्रकियाओं के निमााण 
पर रहा है, तात्कालिक निगरािी और अविरत प्रकियाओं के नियंत्रण पर सीलमत ध्याि टदया 
गया है। इस थीलसस का उद्देश्य बायोथेरेप्यूटिक्स के अविरत प्रसंस्करण में किटिकि क्िालििी 
ऐटिब्यूट्स (सीक्यूए) की तात्कालिक निगरािी के लिए उपयुक्त ऑििाइि प्रोसेस एिालिटिकि 
िेक्िोिॉजी (पीएिी) उपकरण के विकास के साथ-साथ डाउिस्िीम शुद्चधकरण चरणों के मॉडि 
आधाररत डडजाइि का विकास और नियंत्रण करिा है।  
 
काया का पहिा उद्देश्य कॉिम की स्स्थनत ि प्रदशाि की गुणात्मक निगरािी और मात्रात्मक 
अिुमािि और अविरत कैप्चर िोमैिोग्राफी के रेस्िि साइस्क्िंग अध्ययि के लिए मशीि िनििंग 
(एमएि) एल्गोररदम के अिुप्रयोगों का पता िगािा था। रेस्िि िागत के कारण मोिोक्िोिि 
एंिीबॉडी शुद्चधकरण में यह प्रकिया महत्िपूणा है। कॉिम अिंडता विच्छेदों का पता िगािे और 
प्रोिीि ‘A’ िोमैिोग्राफी में यील्ड में चगरािि की भविष्यिाणी करिे के लिए उन्ित एमएि 
एल्गोररदम प्रस्तावित ककए गए हैं। विकलसत एल्गोररदम द्िारा कॉिम पैककंग और यील्ड में 
चगरािि में सूक्ष्म पररितािों का पता िगाकर विसंगनतयों की सफितापूिाक भविष्यिाणी की 
गई, स्जसस ेनििारक उपायों के लिए तात्कालिक नियंत्रण निणाय िेिे में मदद लमिी। 
 
काया का दसूरा उद्देश्य, मोिोक्िोिि एंिीबॉडी (एमएबी) के लिए जिरि रेि मॉडि (जीआरएम) 
का उपयोग करके अविरत कैप्चर िोमैिोग्राफी में प्रोिीि ‘A’ रेस्िि के विस्ताररत पुिरुपयोग 
के कारण होिे िािी रेस्िि क्षयि की जटिि पररघििा को मॉडि करिा था। हमारे निष्कर्षा 
बताते हैं कक प्रत्येक चि के साथ पोर और कफल्म विसरण धीमा हो जाता है, स्जसस ेप्रोिीि 
स्थािांतरण दर प्रभावित होती है। यह भी देिा गया कक यद्यवप पोर विसरण दर-नियंत्रक चरण 
था, कफल्म विसरण भी चिों की संख्या के साथ धीमा हो रहा था, स्जससे ठोस अिस्था से 
सतत द्रि अिस्था तक प्रोिीि की विसरण दर धीमी होती गई थी। इस काया में प्रस्तावित 
पद्धनत और निष्कर्षा, रेस्िि के दवूर्षत होिे के पीछे अंतनिाटहत कियाविचध के बारे में उपयोगी 
जािकारी के साथ प्रोिीि ‘A’ रेस्िि के प्रदशाि के बारे में विश्िसिीय अिुमाि करिे में सहायता 
कर सकते हैं। 
 



vi 

 

प्रकिया मॉडलिगं के अिािा, अविरत प्रसंस्करण में उत्पाद की गुणित्ता विशेर्षताओं की तात्कालिक 
निगरािी के लिए उपयुक्त पीएिी उपकरण का निमााण आिश्यक है, स्जसस ेउत्पाद की गुणित्ता 
और स्स्थरता सुनिस्श्चत हो सके। इस उद्देश्य में, एक िागत-कुशि दृस्ष्िकोण प्रस्तावित ककया 
गया है स्जसमें एक पारंपररक ऑफिाइि हाई-परफॉरमेंस लिस्क्िड िोमैिोग्राफी (एचपीएिसी) 
का उपयोग ऑििाइि विश्िेर्षण के लिए 2-तरफा/6-पोिा िाल्ि के साथ ककया जाता है, ताकक 
प्रकिया कॉिम से उत्पाद स्िीम की स्िचालित सैंपलिगं और फै्रक्शिेशि की सुविधा एक साथ 
लमि सके। प्रस्तावित ऑििाइि कॉस्ऩ्िगरेशि की बहुपरकारीता को एग्रीगटे्स और एमएबी के 
चाजा िेररएंट्स के पथृक्करण में इसके उपयोग से सत्यावपत ककया गया है। इसके अिािा, 
िांनछत शुद्धता प्राप्त करिे के लिए पूलिगं को सुविधाजिक बिािे हेतु प्रीडडस्क्िि मॉडलिगं का 
प्रयोग ककया गया। अध्ययि यह दशााता है कक प्रस्तावित ऑििाइि एचपीएिसी कॉस्ऩ्िगरेशि 
को अविरत प्रेपरेटिि िोमैिोग्राफी में पीएिी अिुप्रयोगों के लिए उपयोग ककया जा सकता है। 
 
अंत में, एंड-िू-एंड एकीकृत डाउिस्िीम शुद्चधकरण प्रकिया का मॉडि आधाररत डडिाइि और 
नियंत्रण प्रस्तावित ककया गया, जो अविरत एमएबी प्रोसेलसगं में ठोस पूलिगं निणाय िेिे के 
लिए ऑििाइि एचपीएिसी- पीएिी उपकरण के साथ जोडा गया। एमएबी के लिए एकसमाि 
चाजा िेररएंि संघिि प्राप्त करिे के लिए ठोस पलूिगं निणाय आिश्यक है। इस अध्ययि में, 
हमिे एक डाउिस्िीम प्रकिया के एंड-िू-एंड मॉडलिगं का प्रस्ताि ककया, जो डडस्जिि ट्विि के 
लिए प्रस्तुत ककया गया, जो ऑििाइि एचपीएिसी- पीएिी उपकरण के साथ एकीकृत है, जो 
तात्कालिक पूलिगं निणाय प्रदाि करिे के लिए है ताकक एकसमाि चाजा िेररएंि संघिि हालसि 
की जा सके। कैिायि एक्सचेंज िोमैिोग्राफी (सीईएक्स) एिुएि के लिए प्रारंलभक और अंनतम 
कि बबदं ुके पूिाािुमाि के लिए ऑििाइि एचपीएिसी चाजा िेररएंि डेिा का उपयोग करके 
मॉडि दशाओं को तात्कालिक अपडेि ककया गया, स्जसके अिुसार िांनछत चाजा िेररएंि संघिि 
को प्राप्त करिे के लिए प्रकिया िोमैिोग्राफी को संग्रह से अपलशष्ि मोड में स्स्िच करिे के 
लिए निदेलशत ककया गया। नियंत्रण रणिीनत को फीड में चाजा िेररएंट्स की विस्ततृ रेंज के 
लिए सफितापूिाक िागू ककया गया स्जसमें एलसडडक की रेंज 13-17%, मेि की रेंज 18-23% 
और बेलसक िेररएंि की रेंज 59-68% थी। सभी घििा अध्ययिों िे साबबत ककया कक सीईएक्स 
पूि में चाजा िेररएंट्स का एकसमाि वितरण हुआ और मेि िेररएंि के लिए प्रकिया उत्पादि 
>85% था। इस प्रकार ये पररणाम, जैिप्रकिया सचंािि के डडस्जिि ट्विन्स बिािे और उिके 
प्रकिया नियंत्रण के लिए उपयोगी होिे की तकिीकी संभाििा को सफितापूिाक प्रदलशात करत े
हैं।     
एक विश्िसिीय, दीघाकालिक अविरत प्रकिया को अचधक दक्षता और मजबूती के साथ चिाि े
के लिए, हम विश्िास करते हैं कक इस थीलसस में प्रस्तुत अध्ययि कई मॉडलिगं आधाररत 
तकिीकों के विकास के साथ-साथ जैि-प्रकिया इकाई संचाििों के तात्कालिक मॉनििररगं और 
नियंत्रण के लिए उचचत पीएिी उपकरणों के अिुप्रयोग को बढ़ािा देगा। 
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